Systematic voiding programme in adults with urinary incontinence following acute stroke: the ICONS-II RCT

Caroline Watkins,^{1,2*} Svetlana Tishkovskaya,¹ Chris Brown,¹ Chris Sutton,³ Yvonne Sylvestre Garcia,³ Denise Forshaw,¹ Gordon Prescott,¹ Lois Thomas,² Christine Roffe,⁴ Joanne Booth,⁵ Kina Bennett,⁶ Brenda Roe,⁷ Bruce Hollingsworth,⁸ Ceu Mateus,⁸ David Britt^{9†} and Cliff Panton¹⁰

- ¹Lancashire Clinical Trials Unit, Applied Health Research Hub, University of Central Lancashire, Preston, UK
- ²Faculty of Health and Care, University of Central Lancashire, Preston, UK
 ³Centre for Biostatistics, Division of Population Health, Health Services Research & Primary Care, School of Health Sciences, Faculty of Biology, Medicine and Health, University of Manchester, Manchester, UK
- ⁴Research Institute for Science and Technology in Medicine, Keele University, Stoke-on-Trent, UK
- ⁵Research Centre for Health, School of Health and Life Sciences, Department of Nursing and Community Health, Glasgow Caledonian University, Glasgow, UK
- ⁶Centre for Health Research and Innovation, NIHR Lancashire Clinical Research Facility, Lancashire Teaching Hospitals NHS Foundation Trust, Preston, UK ⁷Faculty of Health, Social Care and Medicine, Edge Hill University, Ormskirk, UK ⁸Division of Health Research, Lancaster University, Lancaster, UK

⁹Chester, UK

¹⁰Manchester, UK

*Corresponding author clwatkins@uclan.ac.uk †In memoriam

Disclosure of interests

Full disclosure of interests: Completed ICMJE forms for all authors, including all related interests, are available in the toolkit on the NIHR Journals Library report publication page at https://doi.org/10.3310/EFTV1270.

Primary conflicts of interest: Kina Bennett is a co-applicant on the National Institute for Health and Care Research (NIHR)-funded i4i MyPad and the NIHR Programme Development for the imPRoving End of life care Practice in stroke cARE (PREPARE) study. Joanne Booth reports another research grant from the NIHR Health Technology Assessment (HTA) programme on urinary incontinence and is a member of the Trial Steering Committee (TSC) of the CATHETER (RCT: CompAring THE clinical

and cosT-Effectiveness of vaRious washout policies versus no washout policy in preventing catheter associated complications in adults living with long-term catheters) study, funded by the HTA programme. Bruce Hollingsworth is a member of the NIHR Professorial Fellowship Committee and reports NIHR funding as co-investigator for the following: the NIHR health and health inequalities impact of a place-based community wealth initiative, Research Design Service North West, School for Public Health Research LiLaC (Liverpool and Lancaster universities Collaboration for public health research), NIHR Collaboration for Leadership in Applied Health Research and Care (CLAHRC) North West Coast and the NIHR Applied Research Collaboration North West Coast. Gordon Prescott reports receiving other grants from NIHR as co-applicant, not related to urinary incontinence research, and is a member of the Data Monitoring Committee (DMC) of the CATHETER trial of preventing catheter-associated complications. Christine Roffe is chairperson of the TSC for the Imaging cerebral Neuro-inflammation in acute and chronic CerebroVascular Disease: a predictor of outcome and biomarker for guiding treatment (IN-CVD) study, funded by the NIHR Efficacy and Mechanism Evaluation (EME) programme; is a member of the DMC for the CAARBS (RCT of: a Calcium channel or Angiotensin converting enzyme inhibitor or Angiotensin Receptor Blocker regime to reduce blood pressure variability following ischaemic Stroke) feasibility study comparing a calcium-channel blocker versus angiotensin-converting enzyme inhibitors/angiotensin receptor blockers-based regime to target blood pressure variability following transient ischaemic attack and minor ischaemic stroke funded by a British Heart Foundation (BHF; London, UK)/The Stroke Association (TSA; London, UK) programme grant (2016-present); is a member of the DMC for the Penumbral Rescue by Normobaric O = O Administration in Patients With Ischaemic Stroke and Target Mismatch ProFile (PROOF) study, examining high-dose oxygen treatment in patients with large vessel occlusion and acute ischaemic stroke, funded by a European Union Horizon 2020 grant (2016–present); is an independent member of the TSC for the LACunar Intervention (LACI-2) Trial-2, funded by a BHF grant; is an independent member of the TSC for the Paramedic Acute Stroke Treatment Assessment (PASTA) study; is a co-applicant member of the TSC for the Right-2 study of pre-hospital treatment with glyceryl trinitrate patches in acute stroke; is a co-applicant member of the TSC for the Tranexamic acid for hyperacute primary IntraCerebral Haemorrhage (TICH-2) study; is a member of the James Lind Alliance Stroke Research Priority Setting Group (2020–present), UK Biotechnology and Biological Sciences Research Council Future Leaders Funding board (2020-present), TSA Research Awards Panel (2019-present), the UK Swallowing Research Group (2019-present) and the UK Stroke Forum Steering Group (2015-present); was a member of the NIHR HTA General Board (2017–21) and the NIHR Stroke National Specialty Group (2015–18); is chairperson of the NIHR Hyperacute Stroke Research Centre Oversight Group (2015–present); and is the Stroke NIHR National Specialty Group Portfolio Lead for Acute Clinical Studies (2015–present). Chris Sutton is a member of the NIHR HTA Commissioning Funding Committee (January 2020-present) and reports receiving institutional funding from NIHR for his role as co-investigator on several projects [not related to stroke or incontinence research, except for funding for the NIHR Global Health Research Group Improving Stroke Care in India (IMPROVISE) to July 2020], was an independent member of the TSC for the ELECtric Tibial nerve stimulation to Reduce Incontinence in Care homes trial and is an independent member of the Programme Steering Committee for the OPTimising Implementation of Ischaemic Stroke Thrombectomy (OPtimIST) programme grant. Lois Thomas was a Health and Social Care Delivery Research (HSDR)-commissioned panel member (2015–19). Caroline Watkins reports several current and recent NIHR-funded projects; is a member of the NIHR advanced fellowship panel, NIHR PARADISE (Predicting af AfteR cArDIac SurgEry - the PARADISE Score: A Clinical Prediction Rule for Post-operative Atrial Fibrillation in Patients Undergoing Cardiac Surgery) oversight panel, international advisory committee of NIHR Global Health Research Group on Atrial Fibrillation and the TSC/DMC for the NIHR Metoclopramide for Avoiding Pneumonia after Stroke (MAPS-2) study; is chairperson of the TSC/DMC for NIHR COVID-NURSE and the DMC for NIHR HomeHealth; is a co-applicant on the NIHR CLAHRC North West Coast Confirming the Mechanism of Motivational Interviewing Therapy after Stroke (COMMITS) project, NIHR Symptoms, Trajectory, Inequalities and Management: Understanding Long COVID to Address and Transform Existing Integrated Care Pathways (STIMULATE-ICP) grant, NIHR PREPARE programme development grant, NIHR Risk rEduction interVEntion for Raised blood preSsurE (REVERSE) Research for Patient Benefit (RfPB) project grant, NIHR Global Health Research Group Stroke In Sierra Leone (SISLE); has two NIHR-funded doctoral fellowships; is chief investigator of the NIHR Global Health Research Group IMPROVISE and IMPROVIng Stroke care in India – Advancing The INSTRUCT Operations and Network (IMPROVISATION); was a deputy director of NIHR CLAHRC North West Coast; and is currently implementation lead of NIHR Applied Research Collaboration North West Coast.

Published July 2022 DOI: 10.3310/EFTV1270

Plain English summary

The ICONS-II RCT Health Technology Assessment 2022; Vol. 26: No. 31 DOI: 10.3310/EFTV1270

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

U rinary incontinence affects around half of stroke survivors. It causes embarrassment and distress, affecting patients' ability to take part in rehabilitation. It also has a major impact on families and may determine whether or not patients are able to return home. Finding the underlying cause and addressing it can prevent, cure or reduce problems. Doing this in a systematic way for everyone with incontinence problems as early as possible after the stroke, while they are still in hospital, may work best. We also wanted to avoid using catheters in the bladder to drain the urine away, as these are often unnecessary and can cause urinary tract infections.

This study aimed to test whether or not continence problems and the use of urinary catheters could be reduced if everyone with incontinence was fully assessed and given the right management and support early after hospital admission. We also wanted to find out if the benefits outweighed the costs.

We planned to involve 1024 men and women with incontinence from 18 stroke units in the study, with 512 people receiving the intervention and 512 receiving usual care. However, the trial was paused because of COVID-19, at which time only 157 participants had been recruited. When we were thinking about restarting the study and looked at its progress, we found that not enough people had agreed to take part and, of those who had agreed, many had not returned their outcome questionnaires. This indicated that the trial was not feasible and should not restart.

We could not make any firm conclusions about whether or not the intervention worked, as not enough people were involved. We found that stays in hospital after stroke are shorter than they were in the past. This suggests that future studies investigating ways of treating incontinence should consider interventions with management and support for incontinence that continue after patients leave the hospital.

Health Technology Assessment

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Impact factor: 4.014

Launched in 1997, *Health Technology Assessment* (HTA) has an impact factor of 4.014 and is ranked 27th (out of 108 titles) in the 'Health Care Sciences & Services' category of the Clarivate 2021 Journal Citation Reports (Science Edition). It is also indexed by MEDLINE, CINAHL (EBSCO), EMBASE, NCBI Bookshelf, DOAJ, Europe PMC, the Cochrane Library, INAHTA, the British Nursing Index, Ulrichsweb, and the Clarivate Science Citation Index.

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

HTA programme

Health Technology Assessment (HTA) research is undertaken where some evidence already exists to show that a technology can be effective and this needs to be compared to the current standard intervention to see which works best. Research can evaluate any intervention used in the treatment, prevention or diagnosis of disease, provided the study outcomes lead to findings that have the potential to be of direct benefit to NHS patients. Technologies in this context mean any method used to promote health; prevent and treat disease; and improve rehabilitation or long-term care. They are not confined to new drugs and include any intervention used in the treatment, prevention or diagnosis of disease.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

This report

The research reported in this issue of the journal was funded by the HTA programme as project number 16/111/31. The contractual start date was in May 2018. The draft report began editorial review in September 2021 and was accepted for publication in February 2022. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health and Care Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the NHS, these of the authors, those of the NHS, the NIHR, the HTA programme or the Department of Health and Social Care.

Copyright © 2022 Watkins *et al.* This work was produced by Watkins *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaption in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HSDR, PGfAR, PHR journals) and Editor-in-Chief of HSDR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Consultant in Public Health, Delta Public Health Consulting Ltd, UK

Dr Peter Davidson Interim Chair of HTA and EME Editorial Board. Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Ms Tara Lamont Senior Adviser, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Dr Catriona McDaid Reader in Trials, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

Professor James Raftery Professor of Health Technology Assessment, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Dr Rob Riemsma Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Professor Helen Roberts Professor of Child Health Research, Child and Adolescent Mental Health, Palliative Care and Paediatrics Unit, Population Policy and Practice Programme, UCL Great Ormond Street Institute of Child Health, London, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk